Author’s response to reviews

Title: Household and community socioeconomic and environmental determinants of child nutritional status in Cameroon

Authors:

Roland Pongou (Roland_Pongou@brown.edu)
Majid Ezzati (mezzati@hsph.harvard.edu)
Joshua A Salomon (jsalomon@hsph.harvard.edu)

Version: 6  Date: 11 April 2006

Author’s response to reviews:

Dear Editor of BMC Public Health,

Enclosed with this cover letter is the final version of our article to be published in BMC Public Health. Changes brought to this version respond to your instructions. To facilitate interpretation of our model, we additionally clarified its specification, distinguishing between individual/household and community level variables (pages 10-11). Our results and interpretation remained unchanged.

Thank you for publishing our work in BMC Public Health.

Sincerely,

Roland Pongou